# Evaluation of Circulating Cathodic Antigen (CCA) Urine-Tests for Diagnosis of *Schistosoma mansoni* Infection in Cameroon

Louis-Albert Tchuem Tchuenté<sup>1,2,3</sup>\*, Césaire Joris Kueté Fouodo<sup>1,4</sup>, Romuald Isaka Kamwa Ngassam<sup>1</sup>, Laurentine Sumo<sup>1</sup>, Calvine Dongmo Noumedem<sup>1</sup>, Christian Mérimé Kenfack<sup>1</sup>, Nestor Feussom Gipwe<sup>1</sup>, Esther Dankoni Nana<sup>1</sup>, J. Russell Stothard<sup>5</sup>, David Rollinson<sup>6</sup>

1 Centre for Schistosomiasis and Parasitology, Yaoundé, Cameroon, 2 Laboratory of Parasitology and Ecology, Faculty of Sciences, University of Yaoundé I, Yaoundé, Cameroon, 3 National Programme for the Control of Schistosomiasis and Intestinal Helminthiasis, Ministry of Public Health, Yaoundé, Cameroon, 4 Université de Yaoundé, IRD UMI 209 UMMISCO, Yaoundé, Cameroon, 5 Disease Control Strategy Group, Liverpool School of Tropical Medicine, Liverpool, United Kingdom, 6 Wolfson Wellcome Biomedical Laboratories, The Natural History Museum, London, United Kingdom

# Abstract

**Background:** The Kato-Katz is the most common diagnostic method for *Schistosoma mansoni* infection. However, the dayto-day variability in host egg-excretion and its low detection sensitivity are major limits for its use in low transmission zones and after widespread chemotherapy. We evaluated the accuracy of circulating cathodic antigen (CCA) urine-assay as a diagnostic tool of *S. mansoni*. In comparison, a low sensitive CCA test (CCA-L) was assessed.

**Methodology:** The study was conducted in three settings: two foci with single *S. mansoni* infections (settings A and B), and one mixed *S. mansoni* – *S. haematobium* focus (setting C). Stool and urine samples were collected from school-children on three consecutive days. Triplicate Kato-Katz readings were performed per stool sample. Each urine sample was tested with one CCA and only the first urine sample was subjected to CCA-L. Urine samples were also examined for *S. haematobium* eggs using the filtration method and for microhaematuria using urine reagent strips. Overall, 625 children provided three stool and three urine samples.

*Principal Findings:* Considering nine Kato-Katz thick smears as 'reference' diagnostic test, the prevalence of *S. mansoni* was 36.2%, 71.8% and 64.0% in settings A, B and C, respectively. The prevalence of *S. haematobium* in setting C was 12.0%. The sensitivities of single Kato-Katz, CCA and CCA-L from the first stool or urine samples were 58%, 82% and 46% in setting A, 56.8%, 82.4% and 68.8% in setting B, and 49.0%, 87.7% and 55.5% in setting C. The respective specificities were 100%, 64.7% and 100%; 100%, 62.3% and 91.3%; and 100%, 42.5% and 92.0%. Mixed infection with *S. haematobium* did not influence the CCA test results for *S. mansoni* diagnosis.

*Conclusions/Significance:* Urine CCA revealed higher sensitivity than CCA-L and triplicate Kato-Katz, and produced similar prevalence as nine Kato-Katz. It seems an attractive method for *S. mansoni* diagnosis.

Citation: Tchuem Tchuenté L-A, Kueté Fouodo CJ, Kamwa Ngassam RI, Sumo L, Dongmo Noumedem C, et al. (2012) Evaluation of Circulating Cathodic Antigen (CCA) Urine-Tests for Diagnosis of *Schistosoma mansoni* Infection in Cameroon. PLoS Negl Trop Dis 6(7): e1758. doi:10.1371/journal.pntd.0001758

Editor: Patrick J. Lammie, Centers for Disease Control and Prevention, United States of America

Received January 15, 2012; Accepted June 12, 2012; Published July 31, 2012

**Copyright:** © 2012 Tchuem Tchuenté et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Funding: This study received financial support from the University of Georgia Research Foundation, Inc., which was funded by the Bill and Melinda Gates Foundation for this SCORE project. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

\* E-mail: tchuemtchuente@schisto.com

# Introduction

Schistosomiasis remains of significant public health importance worldwide, with an estimated 207 million people infected. Over 90% of all schistosomiasis cases are found in sub-Saharan Africa [1,2]. The disease is caused by six species of blood flukes: Schistosoma haematobium, S. mansoni, S. japonicum, S. intercalatum, S. guineensis and S. mekongi. S. haematobium causes urogenital schistosomiasis while the other species cause intestinal disease. The disease affects the poorest of the poor and compromises their development [3]. The development of preventive chemotherapy strategy, the greater access to praziquantel and the increased resources for control have amplified treatment possibilities to the majority of people who need it [4]. Significant progress has been made in the control of schistosomiasis within the past ten years, though this falls short of the target set by World Health Assembly Resolution 54.19 adopted in 2001, which aimed to reach at least 75% of all school-aged children at risk of morbidity by 2010 [5].

In sub-Saharan Africa, *S. haematobium* and *S. mansoni* are the major causes of disease. *S. haematobium* infections can be diagnosed by several approaches, including detection of schistosome eggs in urines and rapid tests such as urine reagent strips for detection of microhaematuria [6]. On the contrary, there is currently no validated rapid test for *S. mansoni*. Because of its

# **Author Summary**

Recent momentum for the control of neglected tropical diseases has generated a renewed interest in the control of schistosomiasis, resulting in the increase of large scale mass drug administration of praziguantel in several countries. As a consequence, there is a need to develop more sensitive diagnostic tools, especially for Schistosoma mansoni, in order to help targeting control or develop better screening strategies for elimination. We evaluated the commercially available point-of-contact circulating cathodic antigen (CCA) as a potential tool for the screening of S. mansoni infections. Our results showed that urine CCA assay had a better sensitivity for detecting S. mansoni infections than single and triplicate Kato-Katz thick smears. The prevalence obtained with CCA was similar or higher than that obtained with nine Kato-Katz. Neither STH nor S. haematobium infections had any impact on the CCA test results. Although further studies are required for its refinement and validation, the overall results revealed CCA as a promising and easy to use tool for the diagnosis of S. mansoni.

simplicity and relatively low-cost, the Kato-Katz technique [7] is widely used for epidemiological field surveys and is recommended by WHO for mapping to determine programme intervention zones, and for monitoring, evaluation and surveillance of intestinal schistosomiasis control programmes [8]. Though the specificity is very high, the sensitivity of Kato-Katz in single stool sample examination is limited by day-to-day variation in egg excretions leading to measurement error in estimating the presence of infection. This is particularly accentuated in areas with high proportions of light intensity infections [9-11]. The increase of large-scale interventions and repeated mass treatment with praziquantel will significantly reduce the prevalence and intensities of schistosomiasis. As consequence of the increase of low-intensity schistosome infections, more light infections will be often missed if single stool samples are examined by the Kato-Katz method, resulting in high underestimation of infection. Therefore, there is a need to develop and validate more sensitive diagnostic tools for S. mansoni infections.

Within the Schistosomiasis Consortium for Operational Research and Evaluation (SCORE) framework, a multi-country evaluation of the commercially available point-of-contact (POC) circulating cathodic antigen (CCA) was conducted. The study aimed to evaluate the utility of the POC urine-based CCA assay as a survey tool to determine the prevalence of S. mansoni. The sensitivity and specificity of the POC-CCA for diagnosis of S. mansoni were assessed, using the cumulative results of the 9 Kato-Katz thick smear readings as 'reference' diagnostic test. In Cameroon, our study was conducted in three distinct epidemiological settings at different endemic levels; two areas of low and moderate endemicity of S. mansoni, and one mixed infection focus of S. mansoni and S. haematobium. As previous data suggested that POC-CCA testing may result in more children believed to be positive than by Kato-Katz testing, with a portion of these being false positives, the manufacturer developed a less sensitive version of POC-CCA. Therefore, in order to further assess the performance of CCA test, this experimental low sensitive CCA dipstick (designated CCA-L) was also tested in comparison to the commercially available CCA assay (designated CCA).

# Methods

## Ethical statement

The study was approved by the National Ethics Committee of Cameroon (Nr 084/CNE/DNM/09), and was a public health exercise through the Ministry of Health and the Ministry of Education. Stool and urine samples were collected from children in schools with the approval of the administrative authorities, school inspectors, directors and teachers. The objectives of the study were explained to the schoolchildren and to their parents or guardians from whom written informed consent was obtained. Children willing to participate were registered. Each child was assigned an identification number and results were entered in a database and treated confidentially. No identification of any children can be revealed upon publication. All children who participated in the study were treated with praziguantel. Other children were treated during the MDA campaign implemented by the national programme for the control of schistosomiasis and intestinal helminthiasis.

### Study area

Based on previous parasitological data, three epidemiological settings were selected for the study: (i) one setting of low endemicity of single *S. mansoni* transmission, i.e. Yaoundé (setting A) in the Centre region of Cameroon; (ii) one setting of moderate endemicity of single S. mansoni transmission, i.e. Makenene (setting B), Centre region; and (iii) one setting where mixed infections of *S. mansoni* and *S. haematobium* occur, i.e. Njombe (setting C) in the Littoral region. Yaoundé is the capital city of Cameroon. Makenene and Njombe are located at approximately 200 and 325 km from Yaoundé, respectively. Investigations were conducted in public primary schools of Obobogo (3.82489 N, 11.50071 E) in Yaoundé, Baloua (4.88287 N, 10.78953 E) in Makenene, and Kompita (4.57898 N, 9.64848 E) in Njombe.

#### Sampling and data collection

According to the literature, a single Kato-Katz thick smear for diagnosis of *S. mansoni* in low endemicity settings has a sensitivity of only 20–30% [10,12]. However, since our study was to be carried out in both low and moderate endemicity settings, we assumed that a single Kato-Katz thick smear has a maximum sensitivity of 60%. The sensitivity of the CCA test is reported to be 80% or higher [13,14]. Using these sensitivity estimates, a significance level of 5%, and a power of 80%, our sample size of complying children was calculated at 90. Assuming a compliance of 70% for the submission of each of three requested stool samples, the number of children to be included in each study setting was at least 199. To achieve this sample size, we selected a sample size of 250 children per study site.

The study was conducted between December 2010 and January 2011. In each of the three settings, about 250 schoolchildren from the upper classes (age 8–12 years) were enrolled, approximately half boys and half girls. Urine and stool samples were collected from these children over three consecutive days. The samples were collected between 11.00 and 14.00, in 60 mL plastic screw-cap vials, transported to the laboratory and processed the same day. In the laboratory, three Kato-Katz thick smear slides per stool sample, using 41.7 mg templates, were prepared and examined for *S. mansoni* and STH. All urine samples were tested using the CCA assays for diagnosis of *S. mansoni*. In addition, the first urine sample of each child was tested with CCA-L. Both CCA and CCA-L assays were obtained from Rapid Medical Diagnostics (Pretoria, South Africa) and performed at ambient temperature according to

the manufacturer's instructions. Briefly, one drop of urine was added to the circular well of the test cassette and allowed to be absorbed entirely into the specimen pad within the well. Then, one drop of buffer (provided with the kit) was added. The test result was read 20 minutes after adding the buffer. Results were determined in a blinded fashion by at least two individuals. In case the control band did not develop, the test was considered as invalid. Valid tests were scored as negative, trace (weak band) or positive (strong band). Due to the lack of standards designed for this test, the trace and positive results were classified as positive.

In addition to the CCA tests, each urine sample was subjected to a filtration method for detection of *S. haematobium* eggs, and with reagent strips (Teco Diagnostics, USA) for microhaematuria assessment. Each urine sample was agitated to ensure adequate dispersal of eggs, 10 mL of urine were filtered through Nucleopore<sup>®</sup> filter, and filters were examined by microscopy for the presence of schistosome eggs. Schistosome infections were recorded; number of eggs was counted and intensity of infection was calculated and expressed as eggs per 10 mL of urine (eggs/10 mL) for *S. haematobium* or eggs per gram of feces (epg) for *S. mansoni*. For urinalysis, reagent strips were immersed in urine and removed immediately. The strip-results were read between 1–2 minutes by direct comparison of the colour blocks printed on the outside of the bottle label.

# Data analysis

The different data were analyzed by the epidemiological unit of the Centre for Schistosomiasis & Parasitology using appropriate statistical tests and methods. Data were entered in a Microsoft Excel spreadsheet, checked and validated. Statistical analyses were carried out using R software version 2.10.0. Only children with complete data records (i.e. 3 POC-CCA assays, 9 Kato-Katz thick smear readings, 3 reagent strip test results and 3 urine filtrations) were included in the final analysis. Sensitivity and specificity of CCA assays were estimated using 9 Kato-Katz thick smear readings as the reference test. The sensitivity was determined as the percentage of subjects with a positive CCA test among those positive for Kato-Katz; and the specificity was defined as the proportion of true negatives, i.e. the percentage of subjects negative for CCA among people negative for Kato-Katz. Positive predictive value (PPV) and negative predictive value (NPV) were also calculated for the different tests.

Analyses were performed for the 3 study settings separately to determine differences in the sensitivity of the CCA assays potentially resulting from different S. mansoni endemicity levels (infection prevalence and intensity) and as a function of S. haematobium co-infection. To obtain a standardized measure of infection, the geometric mean infection intensity of S. mansoni, expressed as the number of eggs per gram of stool (EPG), was estimated for the three study cohorts. For each individual, the classification into light (1-99 EPG), moderate (100-399 EPG) and heavy (≥400 EPG) infection intensity was calculated based on the arithmetic mean of EPGs derived from the 9 Kato-Katz thick smear readings. The thresholds are set by the World Health Organization [3]. The strength of agreement between the CCA and the 9 Kato-Katz thick smears for each endemic setting was assessed by kappa statistics ( $\kappa$ ), as follows: k<0 indicating no agreement, k = 0-0.2 indicating poor agreement, k = 0.2-0.4indicating fair agreement, k=0.4-0.6 indicating moderate agreement, k = 0.6-0.8 indicating substantial agreement, and k = 0.8-1 indicating almost perfect agreement [15,16]. Day-today variation of Kato-Katz results was assessed using McNemar test [17].

An ordinal logistic regression approach was performed to assess the correlation between CCA and CCA-L categories and *S. mansoni* egg counts. The geometric mean egg counts of nine Kato-Katz thick smears per stool sample per day served as continuous explanatory variable, whereas the color reaction of the CCA test was considered as categorical outcome. This statistical procedure was also used to assess the association between CCA and CCA-L test results, expressed as binary outcome variable (negative/positive), with *S. haematobium* egg count as continuous explanatory variable and micro-haematuria as categorical explanatory variable.

Non-overlapping 95% confidence intervals (CI) or p-values < 0.05 were considered as statistical significance.

To further measure the discriminating power of the diagnostic CCA tests, receiver operating characteristic (ROC) curves were used to assess the association between sensitivity and specificity of the assays [18]. This allowed representing the variation of proportion of true positive individuals for Kato-Katz and CCA test in function of proportion of individuals negative for Kato-Katz but positive for CCA. The area under the curve (AUC) indicated the probability to identify accurately a true positive case when the result was simultaneously positive and negative for the CCA test, using single, triplicate and 9-Kato-Katz as the reference tests. AUC>0.7 indicates high discriminating power.

# Results

### Study adherence

A total of 765 pupils were registered and included in the study: 258 in setting A, 251 in setting B and 256 in setting C. Of these children registered, 625 (81.7%) aged 7–15 years old (mean 10.7 years) provided all three requested urine and stool samples: 138 (53.5%) in setting A, 245 (97.6%) in setting B and 242 (94.5%) in setting C.

### Prevalence of S. mansoni (Kato-Katz)

Table 1 summarizes the results of schistosomiasis prevalence obtained in the different study settings, as assessed by the different diagnostics approaches. The 95% confidence intervals (95% CI) for prevalence are shown. For the Kato-Katz method, there was a significant increase of infection prevalence with the increase of the number of Kato-Katz thick smear readings. Indeed, the prevalence figures for *S. mansoni* were 21.0%, 23.9% and 36.2% in setting A; 41.0%, 49.0% and 71.8% in setting B; and 31.4%, 44.6% and 64.0% in setting C; with one, three and nine Kato-Katz thick smears, respectively.

For the overall three study settings A, B and C, the prevalence of *S. mansoni* increased from 32.8% with one Kato-Katz to 41.8% with three Kato-Katz, and 61.0% with nine Kato-Katz thick smear readings.

#### CCA test results (Table 1)

Similarly, there was an increase of infection prevalence with the increase of the number of CCA tests. The prevalence of *S. mansoni* infections as determined by urine CCA were 52.2% and 62.3% in setting A, 69.8% and 83.7% in setting B, and 71.5% and 86.4% in setting C for one CCA and three CCA tests, respectively. For the overall three settings A, B and C the *S. mansoni* infection prevalence increased from 66.6% with one CCA to 81.6% with three CCA tests.

For CCA-L, the *S. mansoni* infection prevalence after a single test was 16.7% in setting A, 51.8% in setting B, 38.4% in setting C, and 38.9% in the overall three settings.

	Setting A			Setting B			Setting C			All Setting	js A-B-C	
Diagnostic test	No. of children tested	No. of children positive	% positive (95% CI)	No. of children tested	No. of children positive	% positive (95% CI)	No. of children tested	No. of children positive	% positive (95% CI)	No. of children tested	No. of children positive	% positive (95% CI)
S. mansoni diagnosis												
Nine Kato-Katz	138	50	36.2	245	176	71.8	242	155	64.0	625	381	61.0
			(28.2–44.3)			(66.2–77.5)			(58.0–70.1)			(57.1–64.8)
Three Kato-Katz (stool day 1)	138	33	23.9	245	120	49.0	242	108	44.6	625	261	41.8
			(16.8–31.0)			(42.7–55.2)			(38.4–50.9)			(37.9–45.6)
One Kato-Katz (stool day 1)	138	29	21.0	245	100	41.0	242	76	31.4	625	205	32.8
			(14.2–27.8)			(34.7–47.0)			(25.6–37.3)			(29.1–36.5)
Three CCA	138	96	62.3	245	205	83.7	242	209	86.4	625	510	81.6
			(54.2–70.4)			(79.0–88.3)			(82.0–90.7)			(78.6–84.6)
One CCA (urine day 1)	138	72	52.2	245	171	69.8	242	173	71.5	625	416	66.6
			(43.8–60.5)			(64.0–75.5)			(65.8–77.2)			(62.9–70.3)
One CCA-L (urine day 1)	138	23	16.7	245	127	51.8	242	93	38.4	625	243	38.9
			(10.4–22.9)			(45.6–58.1)			(32.3–44.6)			(35.1–42.7)
S. haematobium diagnosis												
Three urine filtrations	138	0	0	245	0	0	242	29	12.0	625	29	4.6
									(07.9–16.1)			(3.0–6.3)
One urine filtration (urine day 1)	138	0	0	245	0	0	242	9	2.5	625	9	1.0
									(0.5–4.4)			(0.2–1.7)
Three Hemastix test	138	5	3.6	245	20	8.2	242	36	14.9	625	61	9.8
			(0.5–6.7)			(4.7–11.6)			(10.4–19.4)			(7.4–12.1)
One Hemastix test (urine day 1)	138	2	1.4	245	11	4.5	242	24	9.9	625	37	5.9
			(0.0–3.4)			(1.9–7.1)			(6.2–13.7)			(4.1–7.8)

**Table 1.** Prevalence of *S. mansoni* and *S. haematobium* infections according to each diagnostic test and stratified by epidemiological setting.

www.plosntds.org

mansoni infections.
S.
٩
diagnosis (
the
for
techniques
different
between
. Agreement
N
Ð

		Setting A	-			Setting B				Setting C				All setting	js A-B-C		
Diagnostic test	Test result	Nine Kato	o-Katz thick	smears		Nine Kato	-Katz thick	smear	s	Nine Kato	-Katz thick	smear	5	Nine Kato	-Katz thick	mears	
		Positive	Negative	*×	P-value	Positive	Negative	*¥	P-value	Positive	Negative	*	P-value	Positive	Negative	*¥	P-value
One Kato-Katz (stool day 1)	Positive	29	0	0.63	8.8 10 <sup>-16</sup>	100	0	0.42	4.4 10 <sup>-16</sup>	76	0	0.4	3.1 10 <sup>-15</sup>	205	0	0.47	0
	Negative	21	88			76	69			79	87			176	244		
Three Kato-Katz (stool 1)	Positive	33	0	0.71	0	120	0	0.54	0	108	0	0.62	0	261	0	0.62	0
	Negative	17	88			56	69			47	87			120	244		
One CCA (urine day 1)	Positive	41	31	0.42	$1.14 \ 10^{-7}$	145	26	0.43	7.1 10 <sup>-12</sup>	136	37	0.47	$7.6  10^{-14}$	322	94	0.47	0
	Negative	6	57			31	43			19	50			59	150		
Three CCA	Positive	46	40	0.4	$5.84  10^{-8}$	160	45	0.29	9.8 10 <sup>-7</sup>	149	60	0.31	3.4 10 <sup>-9</sup>	355	145	0.37	0
	Negative	4	48			16	24			6	27			26	66		
One CCA-L	Positive	23	0	0.52	3.1 10 <sup>-12</sup>	121	6	0.49	0	86	7	0.41	$3.3  10^{-13}$	230	13	0.5	0
	Negative	27	88			55	63			69	80			151	231		

## Agreement between Kato-Katz and CCA assays

Table 2 shows the agreement between the different diagnostics approaches and the reference test, i.e. nine Kato-Katz thick smears, for the diagnosis of *S. mansoni*, stratified by study settings. There were a substantial agreement between the nine Kato-Katz thick smears and one Kato-Katz in setting A (k = 0.6), and three Kato-Katz in settings A (k = 0.7) and C (k = 0.6). The agreement was moderate with one Kato-Katz in setting B (k = 0.4) and three Kato-Katz in settings B (k = 0.4). Finally, the agreement was fair with three CCA in all three settings (k = 0.3-0.4) and with one Kato-Katz in setting C (k = 0.4).

#### Sensitivity, specificity and predictive values of CCA tests

The sensitivity, specificity, positive predictive value (PPV) and negative predictive value (NPV) of the different diagnostic tests were determined and the results are summarised in Table 3. Using the nine Kato-Katz thick smears as 'reference' diagnostic test, the sensitivity varied from 46% to 96.1%. In all three settings, there was an increase of sensitivity with the increase from single to triplicate Kato-Katz or CCA tests. The lowest sensitivity was obtained with CCA-L in setting A (46%). However, when considering the overall three settings, the lowest sensitivity was obtained with single Kato-Katz. The triplicate CCA produced the highest sensitivity in setting C (96.1%) and for the overall three settings (93.2%), followed by the single CCA (84.5%) and the triplicate Kato-Katz (68.5%).

The specificity of CCA test was higher with CCA-L, reaching 100% in setting A. Apart from the setting C, the specificity of CCA assays increased from single to triplicate tests. For the overall three settings, the specificity increased from 40.6% with triplicate CCA to 61.5% with single CCA and 94.7% with CCA-L.

There was a significant variation of PPV and NPV of CCA and CCA-L between study sites, with a higher PPV value of 100% for CCA-L in setting B and a lower value of 53.5% for triplicate CCA in setting A. The NPV varied from 53.4% for CCA-L in setting B to 92.3% for triplicate CCA in setting A (Table 3).

In addition to the reference nine Kato-Katz, the sensitivity of the different assays was assessed towards the combined results of all tests (nine Kato-Katz, triplicate CCA and single CCA-L). Similar trends of increased sensitivities with the increase of number of tests were obtained (Table 3).

#### Effects of S. mansoni infection intensities

To determine if the CCA and CCA-L results were affected by the intensities of S. mansoni infections, the data were analyzed considering the different intensity classes estimated from the 9 Kato-Katz thick smear readings. The results are summarized in Table 4, and the impact of infection intensities on CCA and CCA-L results is illustrated in Figure 1. For all three assays (triplicate CCA, single CCA and single CCA-L) there was an increase of prevalence and sensitivity with the increase of S. mansoni infection intensities in all three settings, apart from single CCA in setting A. In moderate and heavily infected children, the prevalence of single and triplicate CCA was above 90% in all three settings. High proportions of individuals negative for nine Kato-Katz thick smears were positive for all CCA tests. Indeed, for the overall three settings, the prevalence of positive CCA among Kato-Katz negative individuals varied from 5.3% for CCA-L to 59.4% for triplicate CCA.

	Setting A				Setting B				Setting C				VII settings /	V-B-C		
	Sensitivity	Specificity	ΡΡV	NPV	Sensitivity	Specificity	νqq	NPV	Sensitivity	Specificity	PPV N	PV	ensitivity 3	Specificity	νqq	NPV
	% CI	% CI	% CI	% CI	% CI	%CI	% CI	% CI	% CI	% CI	% CI %	Ū	° CI		% CI	% CI
Nine Kato-Katz as 'reference' diagnostic test																
Single Kato-Katz (stool day 1)	58	100	100	80.7	56.8	100	100	47.6	49.0	100	100 5.	2.4	3.8	00	100	58.1
	(44.3–71.7)	/	/	(73.3-88.1)	(49.5–64.1)	/	/	(39.5–55.7)	(41.2–56.9)		/ 4	4.8–60.0) (	48.8–58.8) /			(53.4–62.8)
Triplicate Kato-Katz (stool day 1)	66	100	100	83.8	68.2	100	100	55.2	69.7	100	100 6	4.9 6	8.5	00	100	57
	(52.9-79.1)	/	/	(76.8–90.9)	(61.3–75.1)	/	/	(46.5–63.9)	(62.4–76.9)		/ (5	6.8-73.0) (	53.8-73.2) ,			(62.2-71.9)
Triplicate CCA	92	54.5	53.5	92.3	90.9	34.8	78.0	60	96.1	69.0	71.3 8	1.8	3.2	9.01	71	79.2
	(84.4–99.5)	(44.1–64.9)	(42.9–64.0)	(85.1–99.6)	(86.7–95.2)	(23.5-46.0)	(72.4–83.7)	(44.8–75.2)	(93.1–99.2)	(59.2–78.7)	(65.2–77.4) (6	8.7–95.0) (	90.6–95.7) (	34.4-46.7)	(67.0-75.0)	(72.1–86.3)
Single CCA (urine day 1)	82	64.7	56.9	86.4	82.4	62.3	84.8	58.1	87.7	42.5	78.6 7.	2.5 8	4.5	51.5	77.4	71.8
	(71.6–92.6)	(54.8-74.8)	(45.5–68.4)	(78.1–94.6)	(76.8–88.0)	(50.9–73.8)	(79.4–90.2)	(46.9–69.3)	(82.6–92.9)	(32.1–52.9)	(72.5–84.7) (6	1.9-83.0) (	30.9-88.1) (	55.4-67.6)	(73.4-81.4)	(65.7–77.9)
CCA-L	46	100	100	76.5	68.8	91.3	95.3	53.4	55.5	92.0	92.5 5.	3.7 6	0.4	94.7	94.7	50.5
	(32.2–59.8)	/	/	(63.8-84.3)	(61.9–75.6)	(84.7–98.0)	(91.6–99.0)	(44.4–62.4)	(47.7–63.3)	(86.2–97.7)	(87.1–97.8) (4	5.7-61.7) (	55.5-65.3) (	91.9–97.5)	(91.8–97.5)	(55.6–65.4)
Combined results as 'reference' diagnostic test	10															
Single Kato-Katz (stool day 1)	32.2	100	100	44.0	45.2	100	100	16.6	35.3	100	100		6	00	100	23.6
	(22.6–41.9)	/	/	(34.7–53.4)	(38.7–51.8)	/	/	(10.5–22.6)	(29.0–41.7)		/ (1	0.7–21.9) (	34.8-43.1)			(19.5–27.6)
Triplicate Kato-Katz (stool day 1)	36.7	100	100	45.7	45.3	100	100	19.2	50.2	100	100 2	1.0	2.6	00	100	27.2
	(26.7–46.6)	/	/	(36.2–55.2)	(47.7–60.9)	/	/	(12.3–26.1)	(43.5–56.9)		/ (1	3.4–26.9) (	45.3–53.9)			(22.6–31.8)
Single CCA (urine day 1)	80	100	100	92.3	77.4	100	100	60	80.5	100	100 8	8.	9.1	00	100	79.2
	(71.7–88.3)	/	/	(85.1–99.6)	(71.9–82.9)	/	/	(44.8–75.2)	(75.2–85.8)	/	/ (6	8.7–95.0) (	75.6–82.6)		/	(72.1–86.3)
Triplicate CCA	95.6	100	100	72.7	92.8	100	100	32.4	97.2	100	100 3	9.1 9.1	5.1	00	100	47.4
	(91.3–99.8)	/	/	(62.0-83.5)	(89.3–96.2)	/	/	(21.8–43.1)	(95.0–99.4)	/	/ (3	(2.6–50.6) (	93.2–96.9)			(40.6–54.1)
CCA-L	25.6	100	100	41.7	57.5	100	100	20.3	43.3	100	100 1	3.1 2	6.2	00	100	25.9
	(16.5–34.6)	/	/	(32.7–50.8)	(50.9–64.0)	/	/	(13.1–27.6)	(36.6–49.9)	/	/ (1	1.9–24.3) (	41.9–50.5)			(21.5–30.3)
doi:10.1371/journal	l.pntd.000175	3.t003														

Table 4. Correlation between the prevalence of CCA test and the intensities of S. mansoni infections.

Diagnostic test	Intensity of S. mansoni	Setting A	Setting B	Setting C	All Settings A-B-C
		% positive (95% CI)			
Three CCA	Negative	45.5 (35.1–55.9)	65.2 (54.0–76.5)	69.0 (59.2–78.7)	59.4 (53.3–65.6)
	Low	76.5 (56.3–96.6)	71.4 (58.8–84.1)	91.8 (84.9–98.7)	81.9 (75.2–88.6)
	Moderate	100	97.3 (92.1–100)	97.1 (91.6–100)	97.5 (94.2–100)
	Heavy	100	98.9 (96.7–100)	100	99.4 (98.3–100)
One CCA (urine day 1)	Negative	35.2 (25.2–45.2)	37.7 (26.2–49.1)	42.5 (32.1–52.9)	38.5 (32.4–44.6)
	Low	84.7 (42.0-87.4)	50.0 (37.0–65.0)	75.4 (64.6–86.2)	64.6(56.2–72.9)
	Moderate	77.8 (50.6–100)	91.9 (83.1–100)	94.3 (86.6–100)	91.4 (85.2–97.5)
	Heavy	95.8 (87.8–100)	95.6 (91.3–99.8)	96.6 (92.0–100)	96.0 (93.0–98.9)
One CCA-L (urine day 1)	Negative	0	08.7 (2.1–15.4)	08.0 (2.3–13.8)	5.3 (2.5–8.1)
	Low	17.6 (00–35.8)	16.3 (6.0–26.7)	23.0 (12.4–35.5)	19.7 (12.8–26.6)
	Moderate	33.3 (2.5–64.1)	73.0 (58.7–87.3)	57.1 (40.7–73.5)	39.4 (30.9–47.9)
	Heavy	70.8 (52.6–89.0)	95.6 (91.3–99.8)	88.1 (79.9–96.4)	89.6 (85.0–94.1)

doi:10.1371/journal.pntd.0001758.t004

# Effects of *S. haematobium* infections and microhaematuria

In setting C, the prevalence of *S. haematobium* increased from 2.5% with one urine filtration to 12.0% with three filtrations. Similarly, the prevalence of micro-haematuria increased from 9.9% after one test to 14.9% after three tests on three consecutive days (Table 1). Interestingly, micro-haematuria was also found in settings A (3.6%) and B (8.2%), non endemic for *S. haematobium* and where all urine filtrations were negative. These 25 children with micro-haematuria were aged between 7–14 years. They were mainly girls, with only 4 boys.

A logistic regression analysis and adjustments were used to assess the correlation between CCA test categories and *S. haematobium* egg counts or microhaematuria. The results showed the absence of correlation between the CCA and CCA-L positivity and the concurrent infection with *S. haematobium*. OR = 0.9 (p = 0.10) for triple CCA, 0.9 (p = 0.16) for single CCA and 0.87 (p = 0.22) for CCA-L. Similarly, there was no significant association between the CCA or CCA-L positivity and the presence of micro-haematuria (p > 0.05).

# Effects of STH infections

In order to assess the potential effects of infections with Ascaris lumbricoides, Trichuris trichiura or hookworms, the CCA and CCA-L responses was compared between individuals non infected and infected with any of these three species of soil-transmitted helminths (STH). The results show no significant difference of percentages of positive CCA and CCA-L based on STH infections status of individuals; 54.2% vs 62.1% for triple CCA, 36.1% vs 39.7% for single CCA and 6% vs 5% for CCA-L, for STH positive and negative individuals, respectively. This suggests that STH infections do not influence the urine CCA assay results.

# Day to day variation of CCA tests

Detailed analysis of the data showed a day-to-day variation of the results of CCA in 21 children: 5 in setting A, 8 in setting B and 8 in setting C. It appears that in these few 21 individuals (16 positive for Kato-Katz and 5 negative) the CCA test varied from positive to negative from one day another, and *vice versa*, even in children positive for Kato-Katz, i.e. infected with S. mansoni.

#### Discriminating power of CCA tests: ROC curves and AUC

To assess the discriminating power of CCA tests, data of all three settings were merged together. The ROC curves and AUC of the different CCA and CCA-L assays are presented in Figure 2. It appears that the discriminating powers of single CCA and CCA-L were high in all cases, as the values of AUC were >0.7. However, triplicate CCA was less discriminating, with values of AUC (0.6-0.7) < 0.7.

#### Discussion

The urine-CCA cassette test is a rapid antigen detecting test of active schistosome infection, more specifically *S. mansoni*. As the gastrointestinal tract of schistosome is a cul-de-sac, the parasite has to regurgitate at regular intervals the undigested particulate material as well as parasitic gut associated glycoproteins. CCA is one of the major antigens regurgitated by adult schistosomes and the major portion is later secreted in the host urine. The results of our study showed that urine CCA tests gave higher prevalences of *S. mansoni* than single and multiple Kato-Katz thick smears in all three investigated settings. This higher sensitivity of CCA is consistent with previous studies [19,20]. However, detailed analysis of data revealed the complexity of results.

In the three epidemiological settings, the *S. mansoni* prevalence increased by about twofold when moving from single to nine Kato-Katz thick smears. This result confirms the low sensitivity of single Kato-Katz and its improvement with the increase of the number of thick smears [21–23]. Taking into account the fact that repeating Kato-Katz allows a better estimation of the 'true' prevalence of schistosome infection, we adopted a  $3 \times 3$  Kato-Katz design, i.e. 3 stool samples collected per child on three consecutive days and 3 Kato-Katz readings per sample; and the cumulative results of the 9 Kato-Katz thick smear readings was used as our 'reference' diagnostic test. The sensitivity and specificity of the POC-CCA and CCA-L assays were determined by comparison to this reference test. The sensitivity of the normal CCA was high (>82%) in all three settings, whereas the specificity was low, ranging from 42.5% to 64.7%. On the contrary, the sensitivity of









**Figure 1. Correlation between** *S. mansoni* **infection intensities and CCA results.** The prevalences of CCA (single CCA, triple CCA and CCA-L) are affected by the abundance of *S. mansoni* infection (stratified by intensity classes). A = setting 1, B = setting 2, C = setting 3, D = All settings A-B-C. doi:10.1371/journal.pntd.0001758.g001

CCA-L was low (overall 60%) and the specificity was quite high, above 91% in settings B and C, and up to 100% in setting A. This showed the poor accuracy of CCA-L in detecting *S. mansoni* infections. Therefore, this assay, in its current formulation, cannot be recommended for *S. mansoni* diagnosis. Similar results were obtained by Coulibaly et al. [23]. In all three settings, and for the overall settings, a single CCA showed higher or similar high prevalence than nine Kato-Katz thick smears. This suggests that the current commercially available CCA urine cassette assay maybe an appropriate test for the diagnosis of *S. mansoni* in moderate transmission zones, as reported by previous studies in different settings [20,24].

Detailed analysis of our data showed a strong correlation between the *S. mansoni* infection intensities and the positivity rate of CCA, as well as the intensity of CCA test bands. In all settings, CCA prevalences were above 95% and up to 100% in children with high burden of infections (infection intensity  $\geq$ 400 epg), whereas they were only about 65% in light infected children (Figure 1). This high variation of the sensitivity of CCA tests with the intensity of infections is in line with previous studies [19,20,25,26]. Interestingly, our studies showed no influence of STH infections on the CCA results. Furthermore, there was no cross reactivity of *S. haematobium* infections nor microhaematuria (determined by urinalysis reagent strips) on the CCA test results.



**Figure 2. ROC curves and AUC of CCA results using Kato-Katz as reference tests.** The receiver operating characteristic (ROC) curves and the area under the curve (AUC) of CCA and CCA-L are presented using either single (A), triplicate (B) and nine Kato-Katz (C) thick smears as the reference tests. The discriminating powers of single CCA and CCA-L were high in all cases.

doi:10.1371/journal.pntd.0001758.g002

Similar results were obtained in recent studies in Kenya [20] and Côte d'Ivoire [23].

Although far less than for Kato-Katz, day-to-day fluctuations of CCA results were observed in few children all three settings. This daily variation of CCA results from positive to negative and vice versa were observed in 3.4% (21/625) of children. From these 21 children with day-to-day fluctuation in urine CCA, 4 had positive Kato-Katz for all stools collected on the three consecutive days, 6 were negative for all nine Kato-Katz, and 11 exhibited day-to-day variation in Kato-Katz results. Apart from two children with moderate infection intensities, the daily fluctuation of CCA was observed mainly in children with very light infection, with an overall mean S. mansoni egg counts of 1.7 epg. This variation of CCA test results might be linked to the day-to-day fluctuation of schistosome CCA levels in urine of humans infected with S. mansoni, as previously demonstrated in several countries [27-29]. Nevertheless, the fact that CCA is negative in those infected children with egg excretion detected by Kato-Katz raised concerns about the accuracy of CCA for diagnosing S. mansoni infections. According to the manufacturer's leaflet, false CCA negative during the parasitic developing phase, usually in the first 4-8 weeks after infection. However, once adult worms start laying eggs, as detected by Kato-Katz, it becomes difficult to explain false CCA negative results without recourse to more detailed physiological examinations of the worms themselves. Further studies in the laboratory, perhaps, maybe required to address this issue.

The comparison of single CCA and triplicate CCA results showed that the sensitivity of CCA increases with repetition of tests, whereas the specificity decreases. This is confirmed by the value the value of AUC (<70%). This indicates that repeating CCA tests significantly increases the number of false positive individuals to the detriment of true positive.

In conclusion, considering the fact that urine CCA cassette produced similar prevalence as nine Kato-Katz thick smears, this assay seems an attractive tool for the diagnosis of *S. mansoni* infections. Moreover, its rapidity, easy to use, less time consuming than Kato-Katz, and the relative easiness to collect urine than stool samples are significant advantages for this test. Its field applicability for large scale screening for *S. mansoni* infections would be very useful in control programmes, especially as we are moving toward schistosomiasis elimination where feasible. However, further improvement of the assay and efforts to reduce the cost of CCA cassettes are required to enable its large scale. On the contrary, the results of the CCA-L dipstick, in its current formulation, were not satisfactory.

# Acknowledgments

We are grateful to Prof Maurice Tchuenté and Dr Anne Béatrice Dufour for their assistance with this work. We would also like to thank the technicians, school headmasters, teachers and all people who contributed in any way for their assistance and help with the collection of information and data necessary for this study.

# **Author Contributions**

Conceived and designed the experiments: LATT JRS DR. Performed the experiments: LATT CJKF RIKN LS CDN CMK NFG EDN. Analyzed the data: LATT CJKF. Contributed reagents/materials/analysis tools: LATT. Wrote the paper: LATT.

## References

- Steinmann P, Keiser J, Bos R, Tanner M, Utzinger J (2006) Schistosomiasis and water resources development: systematic review, meta-analysis, and estimates of people at risk. Lancet Infectious Diseases 6: 411–425.
- Utzinger J, Raso G, Brooker S, de Savigny D, Tanner M, et al. (2009) Schistosomiasis and neglected tropical diseases: towards integrated and sustainable control and a word of caution. Parasitology 136: 1859–1874.
- World Health Organization (2002) Prevention and control of schistosomiasis and soil-transmitted helminthiasis. Report of a WHO Expert Committee. WHO *Technical Report Series* No. 912. World Health Organization, Geneva.
- 4. World Health Organization (2006) Preventive chemotherapy in human helminthiasis: coordinated use of anthelminthic drugs in control interventions: a manual for health professionals and programme managers. World Health Organization Press, Geneva.
- World Health Organization (2011) Schistosomiasis: number of people treated, 2009. Weekly Epidemiological Record 9 (86): 73–80.
- Mott KE, Dixon H, Osei-Tntu E, England EC, Tekle A (1985) Evaluation of reagent strips in urine tests for detection of *Schistosoma haematobium* infection: a comparative study in Ghana and Zambia. Bull WHO 63: 125– 133.
- Katz N, Chaves A, Pellegrino J (1972) A simple device for quantitative stool thick-smear technique in *Schistosomiasis mansoni*. Rev Inst Med Trop Sao Paulo 14: 397–400.
- 8. World Health Organization (1991) Basic Laboratory Methods in Medical Parasitology. World Health Organization, Geneva.
- Hall A (1981) Quantitative variability of nematode egg counts in faeces: a study among rural Kenyans. Trans R Soc Trop Med Hyg 75: 682–687.
- Booth M, Vounatsou P, N'Goran EK, Tanner M, Utzinger J (2003) The influence of sampling effort and the performance of the Kato-Katz technique in diagnosing *Schistosoma mansoni* and hookworm co-infections in rural Côte d'Ivoire. Parasitology 127: 525–531.
- Tarafder MR, Carabin H, Joseph L, Balalong Jr E, Olveda R, et al. (2009) Estimating the sensitivity and specificity of Kato-Katz stool examination technique for detection of hookworms, *Ascaris lumbricoides* and *Trichuris trichiura* infections in humans in the absence of a 'gold standard'. Int J Parasitol 40(4):399-404.
- Raso G, Vounatsou P, McManus DP, N'Goran EK, Utzinger J (2007) A Bayesian approach to estimate the age-specific prevalence of Schistosoma mansoni and implications for schistosomiasis control. Int J Parasitol 37: 1491– 1500.
- van Dam GJ, Wichers JH, Ferreira TM, Ghati D, van Amerongen A, et al. (2004) Diagnosis of schistosomiasis by reagent strip test for detection of circulating cathodic antigen. J Clin Microbiol 42: 5458–5461.
- Legesse M, Erko B (2007) Field-based evaluation of a reagent strip test for diagnosis of Schistosoma mansoni by detecting circulating cathodic antigen in urine before and after chemotherapy. Trans R Soc Trop Med Hyg 101: 668–673.
- Landis JR, Koch GG (1977) The measurement of observer agreement for categorical data. Biometrics 33: 159–174.

- Cohen J (1960) A coefficient of agreement for nominal scales. Educ Psychol Meas 20: 37–46.
- McNemar I (1947) Note on the sampling error of the difference between correlated proportions or percentages. Psychometrika 12: 153–157.
- Daigle JM (2002) L'utilisation des courbes ROC dans l'évaluation des tests diagnostiques de laboratoire Clinique: Application à l'étude de la pneumonite d'hypersensibilité. In Faculté des sciences et de de génie universitaire de Laval.
- Standley CJ, Lwambo NJS, Lange CN, Kariuki HC, Adriko M, et al. (2010) Performance of circulating cathodic antigen (CCA) urine-dipsticks for rapid detection of intestinal schistosomiasis in schoolchildren from shoreline communities of Lake Victoria. Parasites & Vectors 3:7.
- Shane HL, Verani JR, Abudho B, Montgomery SP, Blackstock AJ, et al. (2011) Evaluation of Urine CCA Assays for detection of *Schistosoma mansoni* infection in Western Kenya. PLoS Negl Trop Dis 5(1): e951.
- De Vlas SJ, Gryseels B (1992) Underestimation of Schistosoma mansoni prevalences. Parasitology Today 8: 274–277.
- Lin DD, Liu JX, Liu YM, Hu F, Zhang YY, et al. (2008) Routine Kato-Katz technique underestimates the prevalence of *Schistosoma japonicum*: a case study in an endemic area of the People's Republic of China. Parasit Int 57: 281–286.
- Coulibaly JT, Knopp S, N'Guessan NA, Silue KD, Furst T, et al. (2011) Accuracy of Urine Circulating Cathodic Antigen (CCA) Test for *Schistosoma* mansoni Diagnosis in Different Settings of Côte d'Ivoire. PLoS Negl Trop Dis 5(11): e1384.
- Stothard JR, Sousa-Figuereido JC, Betson M, Adriko M, Arinaitwe M, et al. (2011) Schistosoma mansoni Infections in Young Children: When Are Schistosome Antigens in Urine, Eggs in Stool and Antibodies to Eggs First Detectable? PLoS Negl Trop Dis 5(1): e938.
- Legesse M, Erko B (2008) Field-based evaluation of a reagent strip test for diagnosis of schistosomiasis mansoni by detecting circulating cathodic antigen (CCA) in urine in low endemic area in Ethiopia. Parasite 15: 151–155.
- Stothard JR, Kabatereine NB, Tukahebwa EM, Kazibwe F, Rollinson D, et al. (2006) Use of circulating cathodic antigen (CCA) dipsticks for detection of intestinal and urinary schistosomiasis. Acta Tropica 97: 219–228.
- Disch J, Garcia MMA, Krijger GW, Amorim MN, Katz N, et al. (1997) Daily fluctuation of levels of circulating cathodic antigen in urine of children infected with *Schistosoma mansoni* in Brazil. Transactions of the Royal Society of Tropical Medicine and Hygiene 91: 222–225.
- Van Etten L, Kremsner PG, Krijger FW, Deelder AM (1997) Day-to-day variation of egg output and schistosome circulating antigens in urine of *Schistosoma haematabium*-infected school children from Gabon and follow-up after chemotherapy. American Journal of Tropical Medicine and Hygiene 57: 337– 341.
- Polman K, Engels D, Fathers L, Deelder AM, Gryseels B (1998) Day-to-day fluctuation of schistosome circulating antigen levels in serum and urine of humans infected with *Schistosoma mansoni* in Burundi. American Journal of Tropical Medicine and Hygiene 59: 150–154.